Author's response to reviews

Title: The Optimal Starting Time Of Postoperative Intraperitoneal Mitomycin-C Therapy With Preserved Intestinal Wound Healing

Authors:

Ali Uzunkoy (aliuzunkoy@yahoo.com)
Cengiz Bolukbas (drcengizbolukbas@hotmail.com)
Mehmet Horoz (mehmethoroz@yahoo.com)
Fusun F Bolukbas (fusunbol@yahoo.ca)
Abdurrahim Kocyigit (kocyigit@harran.edu.tr)

Version: 4 Date: 28 January 2005

Author's response to reviews:

Dear Editor,

We noted that we didn't remove the key words section at the main text of revised manuscript. Thus, we aimed to change it resubmission of new version of revised manuscript without key words section.

Yours' Sincerely,
Mehmet Horoz, MD.